Author's response to reviews

Title: Two Melodies in Concert: Mitral and Pulmonary Valve Replacement Late in Repaired Tetralogy of Fallot

Authors:

Zhi Fang (_myst_fox@vip.163.com)
Jia Hu (humanjia@msn.com)
Xianglan Zhu (690578344@qq.com)
Ke Lin (Edwarddian@163.com)

Version: 3
Date: 9 March 2015

Author's response to reviews: see over
March 9, 2015

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Two Melodies in Concert: Mitral and Pulmonary Valve Replacement Late in Repaired Tetralogy of Fallot” (MS: 7104040561429787). Those comments are valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made corrections which we hope to meet with approval. We appreciate the reviewer’s suggestion and editing for English has been done by a native English speaker in the revised manuscript.

Once again, thank you very much for your comments and suggestions.

Sincerely,

Ke Lin